Author's response to reviews

Title: Treatment with gefitinib after erlotinib-induced liver injury: a case report

Authors:

Katsumi Nakatomi (nakatomi@nagasaki-u.ac.jp)
Yoichi Nakamura (yi-nakamu@umin.ac.jp)
Tetsuya Iida (tetsu1@mocha.ocn.ne.jp)
Shigeru Kohno (s-kohno@nagasaki-u.ac.jp)

Version: 5 Date: 6 November 2011

Author's response to reviews: see over
Editor-in-Chief
Journal of Medical Case Reports

RE: MS1555857153591684
‘‘Treatment with gefitinib after erlotinib-induced liver injury: a case report’’ by
Nakatomi, K. et al.

Dear Editor

Thank you very much for your prompt review and editing of our manuscript. We are happy to learn that our paper with appropriate revisions may be published in *Journal of Medical Case Reports*. We modified our manuscript in accordance with the reviewers’ and editors’ comments. These changes are marked as underlined text in the revised manuscript. Please find below our revisions and replies to the reviewers’ comments.

We hope that the revised manuscript is acceptable for publication in *Journal of Medical Case Reports*.

Sincerely,

Yoichi Nakamura, MD & PhD
Phone: +81-(95)-819-7274
Fax: +81-(95)-849-7285
E-mail: yi-nakamu@nagasaki-u.ac.jp
Answers to Editor and Referees

Comment 1:
In accordance with the Editor’s comment, we added new sentences on page 4, line 11 to line 15.

Comment 2 (Reviewer2):
In accordance with the Editor’s and Reviewer2’s comment, we added the chemical structures of gefitinib and erlotinib in Figure 2. In addition, we changed the sentence about the hepatotoxicity on page 6, line 2 to line 3.

Comment 3 (Referee1):
In accordance with the Editor’s and Reviewer1’s comment, we added new sentences on page 6, line 11 to line 18.

In accordance with Reviewer2’s comment, the paper has been checked by a native English-speaking physician editor.